Author’s response to reviews

Title: Passive Hallux Adduction Decreases Lateral Plantar Artery Blood Flow: a Preliminary Study of the Potential Influence of Narrow Toe Box Shoes

Authors:

Julia Jacobs (juliajacobs530@gmail.com)
Sarah Ridge (sarah_ridge@byu.edu)
Dustin Bruening (dabruening@byu.edu)
K Annie Brewerton (lilann85@hotmail.com)
Jayson Gifford (jaysongifford@byu.edu)
Daniel Hoopes (dhoopesmd@gmail.com)
Aaron Wayne Johnson (wayne_johnson@byu.edu)

Version: 2 Date: 01 Oct 2019

Author’s response to reviews:

Thank you for pointing out the use of non-standard abbreviations and allowing us to correct these. We have eliminated the non-standard abbreviations that were suggested and replaced those with the full term. Specifically, we changed all BF abbreviations to blood flow, and PHA to passive hallux adduction, LPA to lateral plantar artery, and change in BF to change in blood flow in both the abstract and main document and figure descriptions. We have tracked these changes.